Author's response to reviews

Title: The facilitating factors and barriers encountered in the adoption of a humanized birth care approach in a highly specialized university affiliated hospital

Authors:

Roxana Behruzi (roxana.behruzi@umontreal.ca)
Marie Hatem (marie.hatem@umontreal.ca)
Lise Goulet (lise.goulet@umontreal.ca)
william Fraser (william.fraser@umontreal.ca)

Version: 2 Date: 31 August 2011

Author's response to reviews: see over
August 30, 2011

Dear Editor

Enclosed for your consideration is an article entitled:

The facilitating factors and barriers encountered in the adoption of a humanized birth care approach in a highly specialized university affiliated hospital

We appreciate your interest to our paper entitled:

Please find enclosed our responses to the reviewer’s comments and the revised paper. We strongly believe that the changes we have made regarding to the comments, have improved the quality of the manuscript very much. We hope this is satisfactory, and please don’t hesitate to contact me if there is anything else we can help with the paper.

I look forward to hearing from you.

Sincerely
Roxana Behruzi, MSc. PhD
Department of Social and Preventive Medicine,
Faculty of Medicine
Université de Montréal

Adresse
1420 boul. Mont-Royal
Outremont, QC, Canada, H2V 4P3
Email : roxana.behruzi@umontreal.ca